Author's response to reviews

Title: An elderly woman with gallbladder torsion: a case report.

Authors:

Jorine Boer (j.boer3@antoniusziekenhuis.nl)
Djamila Boerma (d.boerma@antoniusziekenhuis.nl)
Tammo S de Vries Reilingh (t.reilingh@kpnplanet.nl)

Version: 6 Date: 13 October 2011

Author's response to reviews: see over
September 29, 2011

To: Professor M. Kidd, MD  
   Editor-in-Chief  
   Journal of Medical Case Reports

Dear Prof. Kidd,

Hereby we would like to resubmit the revised version of the manuscript entitled “A gallbladder torsion presenting as acute cholecystitis in an elderly woman”, for publication in Journal of Medical Case Reports. According to your request we have changed the title of the document conform the journal’s style.

We genuinely hope that we have responded to the editor’s comments in a satisfactory manor.

Sincerely,

Jorine Boer, MD  
Department of Surgery  
St. Antonius Hospital  
Postbus 2500  
3430 EM Nieuwegein  
The Netherlands  
Phone: +31 30 609 20 13  
Fax: +31 30 603 65 78  
Email: j.boer3@antoniusziekenhuis.nl

Djamila Boerma, MD, PhD  
Email: d.boerma@antoniusziekenhuis.nl

Tammo S. de Vries Reilingh, MD, PhD  
Email: t.reilingh@kpnplanet.nl